Author's response to reviews

Title: Antispasmodic and antidiarrhoeal activity of the fruit of Rosa moschata (J)

Authors:

Niaz Ali Dr (niazpharmacist@yahoo.com)
Hina Alam (pharmacist334@yahoo.com)
Aslam Khan Dr (aslamkhan_mkd@yahoo.com.uk)
Ghayour Ahmad (pharmacist@gmail.com)
Wadood Ali Shah (wadudalishah@yahoo.com)
Muhammad Nabi (mnabipharmacist@gmail.com)
Muhammad Junaid (juniphdr@gmail.com)

Version: 5  Date: 20 November 2014

Author's response to reviews: see over
Reviewer's report
Title: Antispasmodic and antidiarrhoeal activity of the fruit of Rosa moschata (J)
Version: 4 Date: 4 November 2014
Reviewer: GERALD NGO TEKE
Reviewer's report:
L 162: Remove the quotation marks. ‘post extract/test sample treatment’.

Answer: the quotation is removed as asked.

Reviewer's report
Title: Antispasmodic and antidiarrhoeal activity of the fruit of Rosa moschata (J)
Version: 4 Date: 2 November 2014
Reviewer: Theophine Okoye
Reviewer's report:
Minor Revisions
Lines 350 and 352; the exact number of n should be stated and not 3-5 or 5-7 as stated respectively.
Answer: the exact number is mentioned as asked.

We thank the reviewers for giving their precious time for critical appraisal of the article.
Regards!
Dr. Niaz Ali
Associate professor Pharmacology, KMU. Pakistan.